Author's response to reviews

Title: Association between adherence to calcium-channel blocker and statin medications and likelihood of cardiovascular events among US managed care enrollees

Authors:

Richard H. Chapman (rchapman@us.imshealth.com)
Jason Yeaw (jyeaw@us.imshealth.com)
Craig S. Roberts (craig.roberts@pfizer.com)

Version: 3 Date: 8 June 2010

Author's response to reviews:

Dear Editor:

Thank you for accepting our manuscript. We have made the following requested formatting changes:

-------------------------------

Major revisions

Affiliations: Each affiliation has been written out in full, in the following format: Department, Institute, City, State, Country.

Acknowledgements and Author Disclosures: This has been moved to two separate sections, both at the end of the manuscript only, not on the title page. Author disclosures have been renamed 'Competing Interests'.

Appendix: We have removed the lines from the Appendix and included the information as normal text in the body of the paper (on page 9).

Minor revisions

References: We were asked to please ensure all citations are placed before the full-stop; we are not sure what "full-stop" refers to, so have not made this revision.

References: We have edited the reference list to contain up to 30 authors' names for each citation, and removed the term "et al."

Figures: The figures have been deleted from the main manuscript file, and left only in the upload system.

Figures: We have checkled the formatting of the figures.

Typography: We have checked the manuscript for any typographical errors and
made final corrections or revisions where necessary.

Please let us know if you see any problems, or need anything else from us before publication.

Regards,

Rick Chapman